Author’s response to reviews

Title: Prevalence and correlates of sexual intercourse among sexually active in-school adolescents: an analysis of five sub-Saharan African countries for the adolescent’s sexual health policy implications

Authors:

Festo Shayo (festocasmir@gmail.com)
Mariam Kalomo (hkmariitz@yahoo.com)

Version: 4 Date: 02 Sep 2019

Author’s response to reviews:

Response to the Editor and Reviewer comments

Editor comments:

I'm afraid the quality of the English used throughout your manuscript does not currently meet our requirements, as there are several spelling and grammatical errors throughout (e.g..."). We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.
Response:
We have copyedited the manuscript for the English language.

Reviewer comments:

Anthony Idowu Ajayi, PhD (Reviewer 3):
The authors have addressed all comments except a thorough language editing of the manuscript. I read through and still saw several grammar errors. Given that the journal does not do language editing, I recommend that the authors get a native speaker to proofread their work.
Response:
We have copyedited the manuscript for the English language.